Author’s response to reviews

Title: Anaesthesiology and Cognitive Impairment: A Narrative Review of Current Clinical Literature

Authors:

Jillian BELROSE (Jillian.Belrose@lhsc.on.ca)

Ruediger Noppens (ruediger.noppens@lhsc.on.ca;rnoppens@uwo.ca)

Version: 2 Date: 02 Dec 2019

Author’s response to reviews:

1. Competing interests

-- As a member of the editorial board [Ruediger Noppens] of this journal, in order to ensure transparency, please declare this in the Competing Interests section of the Declarations.

Included this information to the manuscript.

2. Consent for publication

-- Please amend your “Consent for publication” section in your Declarations. Consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. Seeing as this is not applicable to your manuscript please state “Not Applicable” in this section.

Done.

3. Funding

-- In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Included the information.

4. Formatting

-- In order to be in line with journal requirements, please include the following headings in both the Abstract and Main manuscript: Background, Main Body, Conclusions.
Included the headers to the abstract and the manuscript

5. Abbreviations

-- Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

Included a list at the end of the manuscript.

6. Additional files

-- Please remove the response to the reviewers’ comments, and cover letter files from the file inventory as these are no longer needed at this stage of the editorial process.

Done.

7. Clean copy

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Done.